Author's response to reviews

Title: Familial liability, obstetric complications and childhood development abnormalities in early onset schizophrenia: a case control study

Authors:

Francesco Margari (margari.f@psichiat.uniba.it)
Maria G Petruzzelli (giusypetruzzelli@libero.it)
Paola A Lecce (paolalecce@alice.it)
Orlando Todarello (ortod@psichiat.uniba.it)
Andrea De Giacomo (a.degiacomo@psichiat.uniba.it)
Eliana Lucarelli (elianalucarelli@libero.it)
Domenico Martinelli (domenicomartinelli@gmail.com)
Lucia Margari (l.margari@neurol.uniba.it)

Version: 5 Date: 31 March 2011

Author's response to reviews: see over
For the careful attention of Editor
Melissa Norton

Object: Submission of revised manuscript

Dear Editor

I send you the revised manuscript FAMILIAL LIABILITY, OBSTETRIC COMPLICATIONS AND CHILDHOOD DEVELOPMENT ABNORMALITIES IN EARLY ONSET SCHIZOPHRENIA: A CASE-CONTROL STUDY by Margari F. et al, for publication on your Journal as research article.

In accordance with editorial request of formatting changes we revised the manuscript. Follows a point-by-point description of the changes made.

Major revisions
Author list: the list of authors in the manuscript was written exactly as they are in the submission system, both in style and order.

Minor revisions
Title: in the title and in the manuscript file were capitalised only the first word and proper nouns.
Affiliations: the city was included in the affiliation details.
Abbreviations: abbreviations were formatted in to a paragraph.
Authors' contributions: Authors' contributions were formatted into a paragraph.
References: the reference list was completed with all authors for each citation.
Tables: vertical lines were removed from the tables.
Typography: we checked the manuscript for typographical format.
Looking forward to hearing from you, I send my kindest regards.

Best wishes
Lucia Margari